Author’s response to reviews

Title: NBS1 rs2735383 polymorphism is associated with an increased risk of laryngeal carcinoma

Authors:
Xinmei Hu (hxm6231@163.com)
Juan Liao (liaojuan04@163.com)
Huiliu Zhao (zhl200308@163.com)
Feng Chen (cf_527@163.com)
Xuefeng Zhu (zx4651@163.com)
Jiangheng Li (lijhanghengemail@163.com)
Qingqing Nong (nnqq26@163.com)

Version: 2 Date: 14 Dec 2017

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your comments concerning our manuscript. We have studied these comments carefully and have made modifications and corrections accordingly. Here we provide our responses, point by point, to the comments as listed below.

We hope that the manuscript has been revised satisfactorily and will be acceptable for publication.
Best regards,

Yours sincerely,

Qingqing Nong, MD, PhD
Professor of Preventive Medicine
School of Public Health
Guangxi Medical University
Nanning, 530021, P.R.China
E-mail: nnqq@gxmu.edu.cn

Responds to the editor and reviewer’s comments:

Responds to editor’s comments:

1. In order to offer readers a chance to fully assess your methods and findings, your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/

Answer: Thank you for your suggestions. We have asked a native English speaking colleague to help us copyedit the paper. The language errors throughout the whole paper have been corrected in the paper.

2. Please add a 'Conclusion' section after your discussion.

Answer: Thank you for your suggestion. We have added the “Conclusions” section in lines 13-20, page 15.
3. Please visit our submission guidelines and format your article accordingly: https://bmccancer.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Pay particular attention to the 'Declarations' section; you need to include ALL of the following subsections:

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Answer: Thank you for your suggestion. I have formatted my article according to the submission guidelines and added the subsections of “Ethics approval and consent to participate” (lines 9-11, page 16), “consent for publication” (lines 13-14, page 16), “Availability of data and materials” (lines 16-18, page 16), “Competing interests” (lines 20-21, page 16), “Funding” (lines 1-4, page 17), “Authors’ contributions” (lines 6-9, page 17), “Acknowledgements” (lines 11-15, page 17), and “Authors’ information” (lines 17-22, page 17 and line 1, page 18) in the “Declarations” section.

4. Please remove the titles/captions included in the figures and place them at the end of the main manuscript, after the References, in a section ‘Figures’ where you list the following information: * Figure # (e.g. Figure 1), * Title of Figure, and * Caption of Figure.

Answer: Thank you for your suggestion. We have removed the title included in the Figure 1 and placed it at the end of the main manuscript, after the References, in a section “Figures” (lines 28-32, page 20).
Responds to the reviewer’s comments:

Katarzyna Kiwerska (Reviewer 3): Although the significant corrections have been made, there are still striking language mistakes that needs to be improved. Please, show this manuscript to some English "native speaker" to make such corrections - there are still too many mistakes concerning singular and plural variants, the proper use of present or past tense, the proper use of conjunctions and so on.

Answer: Thank you for your suggestions. We have asked a native English speaking colleague to help us copyedit the paper. The language errors throughout the whole paper have been corrected in the paper.